Author's response to reviews

Title: Systematic analysis of the clinical and biochemical characteristics of maternally inherited hypertension in Chinese Han families associated with mitochondrial genome mutations

Authors:

yuqi liu (ametufo980869@163.com)
chao zhu (zhuchao@163.com)
jie yang (yangjie@163.com)
tong yin (yintong@163.com)
jinliao gao (gaojinl@163.com)
zongbin li (lizongb@163.com)
qinghua ma (magingh@163.com)
yang li (liyangbsh@163.com)
minxin guan (guanmx@163.com)
yundai chen (ametufo4290@gmail.com)
xueping wang (wangxq@163.com)
qinglei zhu (zhuqinlei@163.com)

Version: 6 Date: 10 December 2014

Author's response to reviews: see over
Dear editor:

Thank you for your letter. As the two questions in the letter:

1) All the patients were fully informed the study and signed the informed consent to join the study and consent to publish their individual data. All the patients signed the consents in Chinese. If you need the consents, I could send the scanning of these patients’ consents to you. It had been revised in the method section.

2) In the Authors’ Contributions section, the contribution of the other two authors had been added.

Best regards

Yuqi Liu